Author's response to reviews

Title: Primitive Neuroectodermal Tumor (PNET) of the kidney: a case report.

Authors:

Dr Giorgio Pomara (g.pomara@libero.it)
Francesco Cappello (francapp@hotmail.com)
Maria G. Cuttano (mg.cuttano@med.unipi.it)
Francesca Rappa (francapp@hotmail.com)
Dr Girolamo Morelli (g.pomara@libero.it)
Dr Pierantonio Mancini (g.pomara@libero.it)
Cesare Selli (c.selli@dc.med.unipi.it)

Version: 4 Date: 19 Jan 2004

Dear Sirs,
Since the reviewer's comments accepted the manuscript without revision, we re-submit the final version of the paper.
We made the formatting changes you listed for us:
- Affiliation and manuscript sections were changed as suggested
- All unnecessary capitalisations were removed
- In the manuscript text the references were placed before the full stop.

We hope that in the present version the manuscript can be considered suitable for publication.
Hoping to hear from you soon, yours sincerely

Giorgio Pomara MD